

CEP Discussion Paper No 842

December 2007

Productivity and the Sourcing Modes of Multinational Firms: Evidence from French Firm-Level Data

Fabrice Defever and Farid Toubal





Abstract

We investigate the role of a firm's total factor productivity in its decision to import from its affiliates rather than from independent input suppliers. We propose a slightly modified version of the Antràs and Helpman (2004) model. We assume higher fixed costs under outsourcing and a firm-specific production function. We use detailed French firm-level data that provides a geographical breakdown of French firms' import at product level and their sourcing modes in 1999. We find strong empirical support for the theoretical predictions of the model. In particular, high-productivity firms that have a production process intensive in suppliers' inputs source their inputs through independent foreign suppliers.

Keywords: Productivity, Incomplete Contracts, Intra-firm Trade, Outsourcing JEL classifications: F23, F14, L22, L23

This paper was produced as part of the Centre's Globalisation Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

Acknowledgements

We are really grateful to Pol Antràs, Elhanan Helpman, Philippe Martin, Thierry Mayer, Gianmarco Ottaviano, Frédéric Robert-Nicoud and Jean-Marc Siroen for helpful comments.

Fabrice Defever is a Post-Doctoral Fellow at the Centre for Economic Performance, London School of Economics. Farid Toubal is an Assistant Professor at the Paris School of Economics, University of Paris I Panthéon Sorbonne.

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

© F. Defever and F. Toubal, submitted 2007

ISBN 978-0-85328-217-4

1 Introduction

Firms wishing to import intermediate inputs from foreign markets either require a local foreign partner, who acts as an independent input supplier or need to set up their own affiliates. The choice of sourcing modes plays a key role in strategic management decisions and has triggered considerable research effort into international organization of production. In a new body of work, trade theorists have started to bring modern theories of the firm into models of international trade.¹ Seminal contributions include McLaren (2000), Antràs (2003; 2005), Antràs and Helpman (2004) and Grossman and Helpman (2002; 2003, 2005).

Part of this literature focuses on contract incompleteness occurring because firms cannot specify all possible contingencies, in particular when they operate in foreign markets. In Antràs (2003), a final good producer decides whether to source a specific intermediate input through an independent foreign supplier or to integrate it. Possible cost sharing for specific investments leads to a holdup problem. As emphasized by Antràs (2003), a firm's organizational choice depends heavily on the share of intermediate inputs within the industry that requires the engagement of suppliers. In particular, the final-good producer can alleviate the hold-up problem by offering the supplier a larger share of revenue by using outsourcing when the industry is intensive in intermediate inputs produced by the supplier. If the share of inputs that are produced by the final-good producer is large enough, then it should keep the residual rights of control and should integrate the supplier.

An interesting extension of Antràs (2003) is made by Antràs and Helpman (2004) who consider heterogeneity in firm-level productivity. As in Melitz

¹See Spencer (2005), and Helpman (2006) for detailed surveys of the literature.

(2003), the final-good producer draws its productivity randomly from a common distribution and decides to produce only after paying a fixed cost of entry. Antràs and Helpman (2004) assume that the fixed cost of production abroad is lower by outsourcing than with vertical integration. Since the most productive firms can compensate for the foreign affiliate set-up costs, the most productive firms in the headquarter-intensive sector will source their inputs internally. The Antràs and Helpman (2004) model could well explain why the share of U.S. intra-firm import decreases from 43.7% in 1990 to 39.4% 1999. The assumption concerning the ranking of fixed costs is crucial; reversing it leads to opposite findings (Grossman *et al.*, 2005). It might explain the increase of French intra-firm trade from 19% in 1990 to 35.8% in 1998.

When contracts are incomplete, cross-countries differences in the contracting environment might also explain the prevalence of sourcing modes. The idea that the judicial quality of a country influences the hold-up problem and the value of specific investment was first developed by Nunn (2007). Thus, goods requiring substantial relationship-specific investments tend to be produced in countries with good contract-enforcement institutions. This concept is applied to the organizational choice by Antràs and Helpman (2008) who show that final-good producers and suppliers will be unwilling to make relationshipspecific investments in countries with poor contract enforcement institutions.

So far, empirical evidence using firm-level data on outsourcing is scarce. Tomiura (2007) uses detailed data on Japanese firms and shows that Japanese multinational firms that export through their affiliates are more productive than firms that outsource. This productivity ordering is robust even when firm size, factor intensity, and/or industry are controlled. In a related paper, Tomiura (2005), shows how firms tend to outsource more of their activities overseas when their productivity is higher - or when their products are more laborintensive. Using French firm-level data, Raspiller and Sillard (2004) show that French multinational firms trade intermediate inputs through their affiliates when their production process uses advertising intensively.

This study examines the role of a firm's productivity and of its intensity in suppliers' input for its organizational decision. We build on a theoretical model that incorporates the main features of Antràs and Helpman (2004) and use French-level data. We slightly change the model by introducing a production function that allows both the productivity and the cost share of input provided by suppliers, to vary across firms. In addition, we consider in particular the case when the supplier's input is produced abroad. We assume contrary to Antràs and Helpman (2004) that fixed costs associated to outsourcing are higher than those under vertical integration. Contrary to U.S. firms, French firms perceive outsourcing to be related to higher fixed costs (SESSI survey, 1999). At least 70% of French firms think that trading with an affiliate reduces organization costs. According to Williamson (1985), vertical integration reduces fixed costs because it amalgamates the coordination costs of two organizations. An implication of the Antràs and Helpman (2004) sorting result is that industries with greater dispersion in productivity should feature a larger share of intra-firm imports conducted by multinational firms. This prediction is confirmed by the empirical works of Yeaple (2006) and Bernard *et al.* (2007). Applying his methodology to the French data, we find a negative correlation between productivity dispersion within sector and the share of intra-firm import.² This results suggests a reverse ranking of fixed costs for French firms. Our theoretical model gives three important predictions for the organizational choice that faces final-good producers. Assuming a higher fixed-cost under outsourcing, we show that firms outsource when (i) they intensively use suppliers'

 $^{^2\,{\}rm The}$ results are available in Table D.1 of Appendix D.

inputs in their production process, (ii) they simultaneously are highly productive and intensively use suppliers' inputs in their production process, and (iii) they are highly productive and import inputs from countries that have sound contracting environments.

This paper makes two specific contributions to the related empirical literature. First, we use a new data set at firm-level which contains superior information compared to previous studies. Second, our empirical specification follows directly from theory; especially in the case of the intensity in suppliers' inputs. We use detailed French firm-level data to estimate the predictions of the model. The database provides a geographical breakdown of French firms' imports at product level (HS4) and their sourcing modes – through independent suppliers and/or affiliates – in 1999. This data also includes information that allows a distinction to be made between intermediate inputs and final goods. The survey, which has been carried out by the SESSI, includes French firms trading more than 1 million Euro and that are owned by manufacturing groups that control at least 50% of the equity capital of their foreign affiliate. These limitations sharply reduce the number of participants. However, the coverage remains significant. The survey covers 55% of French total imports and 61% of French total exports. In the present analysis, we focus on 2619 French importers which carry out 72391 transactions. However, the survey provides little information at firm-level. We retrieve this information from the EAEdatabase on the balance sheet and income statement of all firms located in France that have more than 20 employees.³ The data is annual from 1996 to 2002.

The previous literature has approximated the parameter of the headquarter intensity in the Cobb-Douglas production function by sector-specific variables

³ EAE: Enquête Annuelle d'Entreprises.

such as capital and/or skill intensities (Antràs, 2003; Bernard *et al.* (2007); Nunn and Trefler, 2008; Yeaple, 2006). We propose a measure of the intensity in suppliers' input which is directly mapped from theory: it is the total use of intermediate inputs from suppliers in the French firm production process. This measure has the main advantage of being firm-specific and to be directly related to the cost share of input provided by suppliers of the Cobb-Douglas production function.

Contrary to the previous, related, literature, we have estimated a firm's total factor productivity (TFP) using the Olley and Pakes (1996) methodology because it has two main advantages over the standard measures of productivity, such as value-added over employment. First, this methodology takes into account the selection bias that arises because we only observe firms that are included in the survey and which are likely to be the most productive. Second, it solves the endogeneity problem that arises because inputs are chosen by firms according to their productivity.

To test the prediction of the model, our empirical specifications take into account the non linear interaction term between the productivity, the intensity in suppliers' input and the contracting environment. We follow Ai and Norton (2003) and use their estimation methodology that allows interpreting multiplicative terms in logit models.

In line with the theoretical prediction, we find that firms that have a production process intensive in suppliers' inputs do source through independent suppliers. Moreover, we find that the most productive firms source their inputs through independent suppliers. The higher the intensity of the production process in suppliers' inputs, the larger the effect. Contrary to our theoretical prediction, we find that a better contracting environment increases the likelihood of importing inputs from affiliates. This result is likely to be in line with the theoretical prediction of the Antràs and Helpman (2008) model that allows inputs to be partially contractible. They show that if a better contracting environment reduces the non-contractible share of the suppliers' input, then the incentives required by the headquarter will increase.

The remainder of this paper is structured as follows. Section 2 provides the theoretical background and the testable implications of the model. Section 3 presents the data and discusses the empirical strategy. Section 4 proposes a first look at the data. Section 5 presents our main results and provides some robustness checks. Section 6 concludes.

2 Theoretical Background

In this section, we slightly modify the Antràs and Helpman (2004) model and review its core testable implication for firm-level data analysis. We denote by v a vertically integrated firm that sources inputs abroad through its affiliate. We use the subscript o for a firm that sources inputs abroad through an independent supplier.⁴

2.1 Set-up

Consumers are assumed to share Dixit-Stiglitz preferences for differentiated products which generate the inverse demand function $p_j(i) = D_j x_j(i)^{\alpha-1}$ for variety *i* in sector *j*. $p_j(i)$ is the price of this variety, $x_j(i)$ is the quantity demanded, D_j is an index of total demand for the output of sector *j*, and the elasticity of demand is equal to $1/(1-\alpha)$ and is larger than one. All final goods are freely traded with zero transport costs, such that D_j measures world

⁴ As we consider only the case where the supplier's input is produced abroad, it should read "vertical integration offshore" and "outsourcing offshore" instead. Offshoring means that the production of the inputs occurs in a *foreign country*.

demand for the output of sector j.

Each sector j produces a differentiated good under monopolistic-competition. The production of the final good requires the use of two specialized intermediate inputs, x_h and x_m . x_h is produced locally by a headquarter, HQ, with a wage that is normalized to one. x_m is sourced from supplier, M, located in foreign country, l, where the wages is $w^l < 1$. Throughout this paper, we rule out the possibility of sourcing x_m from a national supplier and focus on an international fragmented production process. As in Antràs and Helpman (2004), we assume the output of variety i to be a Cobb-Douglas function of these inputs:

$$Q_i = \theta \left[\frac{x_h}{1-z} \right]^{1-z} \left[\frac{x_m}{z} \right]^z \qquad 0 < z < 1 \tag{1}$$

 θ is the firm-specific productivity parameter. (1-z) is the "intensity in headquarter services". We depart from Antràs and Helpman (2004) and assume that z is firm-specific. A higher z is associated with a more intensive use of the intermediate input from supplier M in production. z thus represents the intensity in supplier's input.

Upon paying a fixed cost f_k - which we label in national units of labor and observing its productivity, θ , the headquarter, HQ, faces a choice when sourcing its input. It can decide to import inputs from an independent supplier and pay a fixed cost f_o or import them from its affiliate and pay a fixed cost f_v . We assume that sourcing through an independent supplier generates a higher fixed cost than sourcing through an affiliate because vertical integration creates economies of scope in the management of diverse activities (Antràs and Helpman, 2004).

The transaction between HQ and M involves incomplete contracts because,

ex-ante the headquarter and the supplier cannot sign enforceable contracts specifying the purchase of specialized intermediate inputs for a certain price. They also cannot observe ex-ante the inputs' quality.

Since x_h and x_m are entirely customized and have no value outside the relationship, both firms face a hold-up problem. After the specific investment has been made, there is renegotiation over how the ex-post quasi rents from the relationship will be shared.

Denoted by β , the share of ex-post gain from trade obtained by the headquarter. Following the property-rights approach, the ex-post bargaining will take place both under outsourcing and under vertical integration (Grossman and Hart, 1986 and Hart and Moore, 1990). However, the distribution of surplus is sensitive to the sourcing mode. More precisely, the outside option of the headquarter would be equal to zero under outsourcing, but would be equal to $\delta \in (0, 1)$ under integration. When it chooses to source inputs through its affiliate, the HQ owns the residual rights of control and seizes control of the unit of production. In that case, we assume that the headquarter receives a fraction δ of the amount of x_m , which translates into an outside option of $\delta^{\alpha} R_v$ assuming CES preferences and a constant markup $1/\alpha$. Along with Antràs and Helpman (2004), we think that δ is related to the protection of property rights or more generally to the contracting environment of the export country.

Given the bargaining framework, the headquarter receives the fraction β_o of the revenue R(i) under outsourcing and the fraction $\beta_v = \delta^{\alpha} + \beta_o(1-\delta^{\alpha})$ under vertical integration, hence $\beta_v > \beta_o$. Once the HQ selects the organization form k, the quantity of intermediate inputs is chosen by M to maximize $(1 - \beta_k)R(i) - w^l x_m$, while the quantity chosen by the HQ to maximize $\beta_k R(i) - w^N x_h$. Thus, on one hand, integration yields the headquarter a higher share of the surplus than under outsourcing. However, on the other hand, the supplier's share of surplus is lower, and the incentives to invest decrease. When choosing their sourcing mode, the headquarter faces a trade-off between having more control and inducing more investment from its supplier.

Ex-ante, the supplier pays a transfer T to the headquarter, which ensures its participation in the relationship which would be equal to its profit.⁵ The choice of ownership is chosen ex-ante by the headquarter to maximize its profit, which includes the transfer. Then, headquarter profit equals:

$$\pi_k^l = D^{\frac{1}{1-\alpha}} \theta^{\alpha/(1-\alpha)} \psi_k^l - f_k^l \tag{2}$$

where

$$\psi_{k}^{l} = \frac{1 - \alpha [\beta_{k}^{l}(1-z) + (1-\beta_{k}^{l})z]}{\left[\frac{1}{\alpha} \left(\frac{1}{\beta_{k}^{l}}\right)^{(1-z)} \left(\frac{w^{l}}{1-\beta_{k}^{l}}\right)^{z}\right]^{\alpha/(1-\alpha)}}$$
(3)

Given its productivity level θ , the final-good producer chooses the mode of sourcing and the location of production that maximizes equation (2). It exits when its productivity level is below a threshold θ , denoted $\underline{\theta}$, because its operating profit is negative. On the other hand, firms with $\theta \geq \underline{\theta}$ stay in the industry and choose their mode of sourcing inputs.

We denote by $\overline{\theta}$, the threshold productivity level of the firm that is indifferent between both sourcing modes. Thus, the firms having a larger productivity level than the productivity threshold $\overline{\theta}$, choose to import inputs from an independent supplier. Firms with a productivity between $\underline{\theta}$ and $\overline{\theta}$ will choose to import inputs from their affiliates. Under the threshold productivity value $\underline{\theta}$, firms exit. Using the free-entry condition, we derive the thresholds of productivity that can be expressed as in Equation (4).

 $^{{}^{5}}$ See Antràs (2003) for details.

$$\underline{\theta} = D^{-1/\alpha} \left[\frac{f_v}{\psi_v(z)} \right]^{(1-\alpha)/\alpha}
\overline{\theta} = D^{-1/\alpha} \left[\frac{(f_o - f_v)}{\psi_o(z) - \psi_v(z)} \right]^{(1-\alpha)/\alpha}$$
(4)

In order to derive the probability that a firm chooses to source its inputs arm-length, we follow Helpman *et al.* (2004) and assume that the producer of variety *i* in sector *j* draws a productivity level θ from a pareto distribution $G(\theta)$ with shape parameter κ .

$$G(\theta) = 1 - \left(\frac{b}{\theta}\right)^{\kappa} \quad for \quad \theta \ge b > 0 \quad , \tag{5}$$

We denote by σ_k the probability that a firm chooses the organization form k. Then, $\sigma_o = [1 - G(\overline{\theta})]/[1 - G(\underline{\theta})]$. Using the pareto distribution and the productivity threshold values, we derive the probability, $\sigma_o = (\underline{\theta}/\overline{\theta})^{\kappa}$, that a firm chooses to source inputs through an independent supplier:

$$\sigma_o = \left[\frac{\psi_o(z)}{\psi_v(z)} - 1\right]^{\kappa(1-\alpha)/\alpha} \left[\frac{f_v}{f_o - f_v}\right]^{\kappa(1-\alpha)/\alpha} \tag{6}$$

As shown in Antràs and Helpman (2004), this organizational choice will mostly depend on the intensity of supplier's input. Under the assumption of higher fixed cost under outsourcing, vertical integration yields higher profit than outsourcing for low enough intensities in supplier's input $(\frac{\psi_o^l}{\psi_v^l} < 1)$, while only the most productive firms will outsource when the intensities in supplier's input are high enough $(\frac{\psi_o^l}{\psi_v^l} > 1)$.

2.2 Testable Implications

The results of this model yield a set of testable predictions concerning the productivity level of the firm, the intensity in supplier's inputs and the contracting environment on the prevalence of each sourcing mode.

2.2.1 Supplier's Input Intensity, z

The supplier's input intensity affects the incentive the final good producer wants to give a supplier. In particular, the more intensive the production is in intermediate inputs that are produced by the supplier, the larger the share of revenue the producer wants to give to the supplier. This is possible under outsourcing where $\beta_O < \beta_V$.

Testable implication 1. The likelihood of sourcing through an independent supplier increases with the supplier's input intensity of the production.

Notice that this result does not depend on the ranking of fixed costs or on the level of a firm's productivity.

2.2.2 Productivity, θ .

In Figure 1, we show simulated levels of the critical productivity value and how they relate to the intensity in supplier's input.⁶





Authors' computation assuming $\alpha = 0.5$, $w_l = 0.1$, $\beta_o = 0.9$, $f_o = 1$, $f_v = 0.2$, $\delta = 0.5$.

⁶ Notice that we carefully choose the value of w_l in order to obtain Figure 1. However, w_l does not affect the ratio $\psi_o(z)/\psi_v(z)$ in equation 6.

Outsourcing appears when a firm has simultaneously a high level of productivity and when its production process is relatively intensive in supplier's inputs. Notice that this result depends on the ranking of fixed costs. If the fixed cost under integration was higher than the one under outsourcing, then high-productivity firms that are intensive in headquarter input prefer vertical integration.

Testable implication 2. The likelihood of sourcing inputs from an independent supplier increases with the supplier's input intensity of the production and with the productivity of the firm.

2.2.3 Contracting Environment, δ_l

The model predicts an indirect relationship between the level of firm's productivity and the contracting environment of the export country. We show in Figure 2 the impact of the firm-level productivity and the contracting environment on the critical values of productivity.

Fig. 2. Firm-level productivity, contracting environment and sourcing modes



Authors' computation assuming $\alpha = 0.5$, $w_l = 0.1$, $\beta_o = 0.9$, $f_o = 1$, $f_v = 0.2$, z = 0.5.

Since $\beta_v = \delta^{\alpha} + \beta_o(1 - \delta^{\alpha})$, a better contracting environment implies a large bargaining power, δ , for the final-good producer. Given the intensity in supplier's input, the final-good producer chooses to internalize for a low level of contracting environment. It gives more incentives to the supplier and chooses to outsource for a high level of contracting environment. Since the fixed cost under outsourcing is higher than under vertical integration, only the most productive final good producer will be able to cover the fixed cost associated to outsourcing.

Testable implication 3. The likelihood of sourcing inputs from an independent supplier increases with the productivity of the firm and the level of contracting environment of the export country.

3 Data and Estimation Strategy

3.1 Data

This paper uses information from a confidential firm-level survey which provides information on the trade organization of French firms in 1999.⁷ The data are provided by *SESSI* (Service des Études Statistiques Industrielles) The survey was addressed to all French firms trading more than 1 million Euro, owned by manufacturing groups that control at least 50% of the equity capital of a foreign affiliate. This limitation sharply reduced the number of firms that answered the survey. However, the coverage remains significant. The data covers 55% of total French imports and 61% of total French exports. We focus on 4249 French importers which carried out 104947 transactions. The survey provides a detailed geographical breakdown of French firms' im-

⁷ Échanges internationaux intra-groupe.

ports and exports at product level (HS4) and their sourcing modes – through independent suppliers and/or affiliates.⁸ A French intra-firm transaction is defined as trade with an affiliate controlled by a single French entity with at least 50% of its equity capital. The *SESSI* defines two types of transaction with independent suppliers: formal contractual relationships that refer to alliances, franchising, joint-ventures and licensing agreements; and informal relationships that involve transactions that use far less detailed contracts.⁹

This survey provides little information at firm level. We retrieve this information from the EAE database. It contains information on the balance sheet and income statement of all firms located in France that have more than 20 employees from 1996 to 2002. The EAE provides firm-level information on sales, capital, labor and intermediates use, fixed assets, as well as the 4-digit $NAF700^{-10}$ sectoral classification of the firm.

3.2 Endogenous Variable: Sourcing Modes

Our first step is to distinguish firms that source their inputs through independent suppliers from those that source their input through affiliates. We use the detailed HS4-digit classification to obtain the information as to whether goods have been imported through affiliates or through independent suppliers. The dependent variable, y_{ijl} , takes the value of one if firm *i* sources input *j* from country *l* through an independent supplier and 0 if the input is imported from an affiliate. We take into account the country dimension because HS4

⁸ A transaction is defined as a specific product imported from a country by a firm. Some transactions were broken into two lines in cases where the firm had to announce an amount higher than the one previously filled by the customs services. We aggregate these lines.

 $^{^{9}}$ 15767 (21.8%) are intra-firm transaction, 45502 (62.9%) are informal outsourcing, 1099 (1.5%) are formal outsourcing and 10023 (13.8%) are mixed strategies (a combination of two sourcing modes).

¹⁰ Nomenclature d'Activité française: nomenclature of French activities.

goods produced in low-income countries are very different from similar goods produced in high income countries (Schott, 2004).

$$y_{ijl} = \begin{cases} 1 \text{ if the transaction has been imported from an independent supplier} \\ 0 \text{ if the transaction has been imported from an affiliate} \end{cases}$$

We restrict our analysis to manufacturing sectors but we do not consider the manufacture of food products, beverages and tobacco because there is no detailed firm-level information for these sectors from the *EAE*. We exclude firms active in the manufacture of coke, refined petroleum products and nuclear fuel since the sourcing modes in this industry are likely to be determined by factors such as national sovereignty (Antràs, 2003). This leaves us with 2619 firms realizing 72391 transactions among which 15767 are imported from affiliates, 46954 are outsourced (formal and informal), and 9670 have been completed using both modes. For our empirical analysis, we exclude firms that import inputs from independent suppliers and from their affiliates simultaneously (mixed strategies).

3.3 Main Explanatory Variables

The model requires a careful approximation of the cost share of input provided by supplier, z. The EEA data set provides information on the total amount of inputs supplied to French firms by independent and affiliated suppliers, irrespective of their location.¹¹ Using this information, we define z as the share of inputs from suppliers in French firms' output:

 $^{^{11}}$ These inputs correspond to the materials and equipment, the small materials and some inputs from general sub-contracting.

$$z = \frac{Input \ from \ suppliers}{Output}$$

We also use the EAE database to estimate firm level total factor productivity. The TFP is estimated as the residual of a three-factor Cobb-Douglas production function, with labor and deflated values of capital and material inputs as production factors. Labor is the firm-specific number of employees. The deflators are obtained from the national accounts system of the French statistical office (INSEE).¹² We estimate the production function using the Olley and Pakes (1996) methodology. The Olley and Pakes methodology controls for the simultaneity bias that arises from the endogeneity of a firm's input selection, which will exist if a firm responds to unobservable productivity shocks by adjusting its input choices. This response yields correlation between the stochastic error term and an explanatory variable in the estimation of the production function, thus leading to a biased OLS estimator. The Olley and Pakes estimator corrects for this possible bias by using the firm's investment decision as a proxy for unobserved productivity shocks. We present this methodology in Appendix A.

We follow Nunn (2007) and capture the variation in contracting environments across countries using the "rule of law" variable for the year 2000 from the *Governance Matters VI* database (Kaufmann *et al.*, 2006). This variable is established on the basis of polls of experts or surveys of businessmen/citizens. It is related to the perceptions of the effectiveness and predictability of the judiciary, and the enforceability of contracts. The variable is configured such that a higher value is associated with a better contracting environment.

¹² Nominal values of output are deflated using two-digit sectoral price indexes. Material inputs are deflated using two-digit sectoral price indexes for intermediate inputs published by the INSEE.

3.3.1 Other Control Variables

The theoretical predictions of the model rely partially on the assumption that the fixed cost of sourcing intermediate inputs through an independent supplier is higher than the fixed cost of sourcing through an affiliate. We include the fixed costs of French firms in our baseline regression. The fixed costs have been defined as total physical assets scaled by sales.

We know from the model that β_o influences the sourcing choice and control for it in the empirical specification by including a firm's size variable. The idea that that firm's size confers a bargaining advantage has received some empirical support (Porter, 1974). The size of the firm is approximated by the number of its employees in 1999.

The organizational choice might also be influenced by country-specific endowment and firm-specific factor intensities. According to Antràs (2003), capital intensive goods are imported from affiliates located in countries that are relatively abundant in capital. We use data on countries' stock physical and human capital and on firm-level factor intensity to examine this relationship. The country-specific endowment data are taken from Trefler (2002). Capital endowment (K/L) is measured by the natural log of the ratio of the physical capital stock divided by the total labor force. Human capital endowment (H/L) is measured by the natural log of workers completing high school to the total labor force. The data on firm factor intensity are taken from the EAE. We use the firm-level capital-labor ratio, k/l, to proxy the firm's capital intensity and its per-employee spending on information technology, s, to roughly control for a firm's skill intensity.

3.4 Estimation Strategy

We analyze the choice between outsourcing and vertical integration using a logit model at the transaction level. From our theoretical framework, the organizational choice is a function of a firm's fixed costs f, a firm's productivity θ , the supplier's input intensity z, the primitive bargaining power β_o and the contracting environment of the exporting country, δ_l . In Equation (7), we denote by subscript i, the firm-specific variable and by subscript l the country-specific variable.

$$y_{ijl} = \lambda_0 + \lambda_1 f_i + \lambda_2 z_i + \lambda_3 \theta_i + \lambda_4 \beta_{oi} + \lambda_5 \delta_l$$

$$+ NAF + HS + \epsilon_{ijl}$$
(7)

where NAF and HS are the sets of sector and product specific fixed effects, respectively. ϵ_{ijl} is the stochastic error term. To estimate the empirical implication of the model, we also need to estimate how the relationship between the intensity in suppliers' inputs and the contracting environment evolves with the productivity. We therefore interact these explanatory variables with the productivity θ .

$$y_{ijl} = \lambda_0 + \lambda_1 f_i + \lambda_2 z_i + \lambda_3 \theta_i + \lambda_4 \beta_{oi} + \lambda_5 \delta_l$$

$$+ \lambda_6 (z_i \times \theta_i) + \lambda_7 (\theta_i \times \delta_l)$$

$$+ NAF + HS + \epsilon_{ijl}$$
(8)

The interpretation of interaction effects in non-linear models, such as logit, is complex. Ai and Norton (2003) argue that odds ratios have no meaningful interpretation for the interaction effects. We follow Ai and Norton (2003) and compute the cross derivative of the expected value of the dependent variable and the statistical significance of the entire cross-derivative. The interaction effect is conditional on other independent variables. Because there are two additive terms, each of which can be positive or negative, the interaction effect may have different signs for different values of covariates.¹³ We discuss the correct interpretation of the interaction effects in the fifth section.

The logit model relies crucially on the assumption of homoskedasticity in the underlying latent variable model. We use the Huber-White method to correct for heteroscedasticity. Finally, because the model is non-linear in its parameters, the marginal effects are not constant and must be interpreted at some sample point. We choose the means of the independent variables for this evaluation. The descriptive statistics and a correlation table are shown in Appendix B (Table B.1 and B.2).

4 A First Look at the Data

4.1 Fixed Costs

The organizational choice depends crucially on ranking of fixed costs. We first have a closer look at this ranking by comparing the cumulative distributions of firms' fixed costs in Figure 3. We only take into account firms that import at least 80% of their total import volumes either through outsourcing agreements, O, or through their affiliates, V.¹⁴ The graph points to a first-order stochastic dominance of outsourcing with respect to fixed costs. Firms that source inputs through independent suppliers have higher fixed costs than firms that import through their affiliates.

We analyze systematic differences between both distributions using the nonparametric Kolmogorov-Smirnov two-sided test (KS-test). The KS-test has

 $^{^{13}}$ We have used the Stata command "inteff" to compute the marginal effect of the interaction terms. See Norton *et al.* (2004).

 $^{^{14}}$ We find qualitatively similar results when we modify the threshold value of 80%, in particular, when firms source entirely through either one of the modes. The drawback of using this latter approach is a loss of 53% in the total number of firms.



Fig. 3. Cumulative distribution of French firms' fixed costs, in logarithm

Source: EAE and SESSI's survey, authors' computation.

the advantage of making no assumption about the sample distribution. It determines if two distributions differ significantly. Therefore, it calculates the largest difference between the observed and expected cumulative frequencies, which is called *D-statistics*. This statistic is compared against the critical D-statistic for that sample size. The results of the KS-test are presented in Table 1.

Table 1 Kolmogorov-Smirnov test for equality of fixed cost distributions

	Difference	P-value	Corrected
$f_o > f_v$	0.0049	0.982	
$f_o < f_v$	-0.0948	0.001	
Combined K-S	0.0948	0.002	0.002

The largest difference between the distribution functions is 0.0948 which is statistically significant at 1%. Thus, the null hypothesis that both fixed costs distributions are equal is rejected. From the left-hand side of the KS test, we can reject the hypothesis that firms that source through their affiliates (V-type) have on average larger fixed costs than firms that source through independent suppliers (*O*-type). The largest difference between the distribution functions is 0.0049, which is not significant. From the right-hand side of the test, we accept the hypothesis that *V*-type firms are less productive than *O*-type firms.¹⁵

4.2 Productivity and the Choice between Sourcing Modes

We look at the distribution of French firms' TFP in Figure 4. The TFP distribution of French firms is not too far from log-normal even if it is slightly right-skewed. It seems that the TFP distribution of French firms does not differ from other firms' TFP distributions (Sutton, 1997; Cabral and Mata, 2003).





Source: EAE and SESSI's survey, authors' computation.

 $^{^{15}}$ Note that increasing the threshold value of 80 % to 100% does not qualitatively change the results.

We have a closer look at firm heterogeneity in Table 2. It reports the number of transactions and the volume of imports according to the productivity distribution of French firms. In particular, it presents four quartiles that correspond to low $(1^{st} \text{ quartile})$, medium $(2^{nd} \text{ quartile})$, high $(3^{rd} \text{ quartile})$, and the most productive firms (4^{th} quartile). Firms in the 4^{th} quartile are on average 1.7% percent more productive than firms in the 1^{st} quartile.¹⁶

Table 2 Quartile distribution of intra-firm trade and outsourcing (percentage of total in parentheses)

Quartile of TFP	Average TFP	Intra-Firm Trade	Outsourcing	Total						
	Number of transactions									
First Quartile	3.67	6682	12812	19494						
		(34.3)	(65.7)							
Second Quartile	4.57	4865	11039	15904						
		(30.6)	(69.4)							
Third Quartile	5.09	6426	13721	20147						
		(31.9)	(68.1)							
Fourth Quartile	6.11	7464	19052	26516						
		(28.1)	(71.9)							
Total	4.86	25437	56624	82061						
	Ir	nport Volume (10	000 Euro)							
First Quartile	3.67	8853.3	8251.0	17104.3						
		(51.8)	(48.2)							
Second Quartile	4.57	5209.0	4910.2	10119.1						
		(51.5)	(48.5)							
Third Quartile	5.09	11774.1	8333.6	20107.7						
		(58.6)	(41.4)							
Fourth Quartile	6.11	11143.1	21404.4	32547.5						
		(34.2)	(65.8)							
Total	4.86	36979.5	42899.2	79878.7						

Given productivity, firms that source through their foreign affiliates trade less

- both in terms of the number and the volume of transactions. We find a

¹⁶ Note that the number of total transactions is slightly higher than the total number of transactions presented in Table 4. This result is mainly driven by firms that have mixed strategies.

	Difference	P-value	Corrected
$TFP_o > TFP_v$	0.0124	0.890	
$TFP_o < TFP_v$	-0.0930	0.001	
Combined K-S	0.0930	0.003	0.002

Table 3 Kolmogorov-Smirnov test for equality of productivity distribution

concentration of activities among the most productive firms.

In order to compare firms' productivity according to their sourcing mode, we compare their cumulative distributions of TFP in Figure 5. The graph points to a first-order stochastic dominance of outsourcing with respect to TFP. Firms that outsource are more productive than firms that import through their affiliates but not over the whole distribution.

Fig. 5. Cumulative distribution of French firm's TFP, in logarithm



Source: EAE and SESSI's survey, authors' computation.

The results of the two-sided KS-tests in Table 3 confirm that the most productive firms import from independent suppliers. The KS-test shows that both distributions of TFPs are statistically different at 1% level of significance. Importantly, the two-sided test rejects the null hypothesis of higher TFPs under integration. However, it accepts the hypothesis that O-firms have higher TFPs than V-firms.

4.3 Suppliers' Input Intensity

Table 4 reports the sector-level aggregates figures concerning the intensity in suppliers' inputs, intra-firm trade and outsourcing. The suppliers' input intensity presented in Table 4 is expressed as the average intensity in suppliers' input within each sector.

For sake of simplicity, we aggregate the NAF700 sectors into 14 industries. We present the number of import transactions and their share as a percentage of the total number of transactions. About 65% of the transactions are exchanged through outsourcing agreements while intra-firm import relates to only 22% of all transactions. The share of transactions that are imported through mixed strategies¹⁷ concerns about 14% of the total number of transactions.

Inter-industry differences with respect to the sourcing modes at transaction level are also apparent in Table 4. Within the six industries with the higher share of outsourcing, four have high relative intensity in suppliers' input: leather products (77.9%), other transport equipment (77.3%), publishing and printing (72.8%) and textile products (73.9%). A higher share of intra-firm import is found in the industries of other non-metallic mineral products (34.7%) and electrical components (32.5%). Both industries are relatively intensive in headquarter services. This observation is in line with the findings of Antràs and Helpman (2004) that firms in headquarter-intensive industries are importing mostly from their affiliates.

 $^{^{17}\,\}mathrm{A}$ combination of both sourcing modes.

Table 4									
Number	of import	transactions	by	sourcing	modes	(percentage	of total	into	paren-
theses)						•			

Industries		Number	Total	Intra-firm	Outsourcing	Mixed
	\widetilde{z}^{\ddagger}	of firms		Import		Strategies
Textiles, textile products	10.4	101	2462	380	1820	262
				(15.4)	(73.9)	(10.6)
Leather, leather products	14.9	66	3650	616	2845	189
				(16.9)	(77.9)	(5.2)
Wood, paper	2.1	165	3617	513	2895	209
				(14.2)	(80.0)	(5.8)
Publishing and printing	12.2	80	886	203	645	38
				(22.9)	(72.8)	(4.3)
Chemicals, rubber and plastic products	5.4	478	13814	3110	8904	1800
				(22.5)	(64.5)	(13.0)
Pharmaceutical products	6.7	168	4701	1026	3234	441
				(21.8)	(68.8)	(9.4)
Other non-metallic mineral products	3.7	128	2843	987	1699	157
				(34.7)	(59.8)	(5.5)
Basic metals and fabricated metal	6.2	314	5653	661	4397	595
				(11.7)	(77.8)	(10.5)
Machinery. equipment n.e.c.	13.0	408	9025	2110	5638	1277
				(23.4)	(62.5)	(14.1)
Electrical component	5.0	157	6177	2010	3140	1027
				(32.5)	(50.8)	(16.6)
Electrical equipment	10.5	206	7206	1806	3905	1495
				(25.1)	(54.2)	(20.7)
Motor vehicles	3.7	125	5279	851	3302	1126
				(16.1)	(62.5)	(21.3)
Other transport equipment	26.3	53	2039	200	1577	262
				(9.8)	(77.3)	(12.8)
Furniture	4.8	170	5039	1294	2953	792
				(25.7)	(58.6)	(15.7)
Total	7.9	2619	72391	15767	46954	9670
				(21.8)	(64.9)	(13.4)

 $\ddagger:\widetilde{z}$ is the average intensity in suppliers' input within each sector

4.4 Contracting Environment and Sourcing Modes

In order to examine the relationship between the contracting environment and the sourcing mode of import, we create a country-specific variable that has information on the share of intra-firm import from a specific country. This variable is defined as the ratio between imports through foreign affiliates to total French imports: Intra-Firm Share $= \frac{M_v}{M_o + M_v}$, where M_v and M_o denote French imports through affiliates and French import through independent suppliers, respectively.¹⁸

 $^{^{18}}$ We exclude countries for which we observe less than 10 transaction. Notice that the positive correlation is not sensitive to this threshold.



Fig. 6. Contracting environment and intra-firm share

Source: EAE and SESSI's survey, authors' computation.

Figure 6 shows that a country-specific contracting environment matters. We find a positive correlation between the intra-firm share and the rule of law variables. This finding is similar to the one reported by Nunn and Trefler (2008) who show that the share of US intra-firm import is positively related to the exporting country's contracting environment. It is however not in line with our theoretical prediction, since a better contracting environment should increase the likelihood of outsourcing given the ranking of French firms' fixed costs. We investigate this result in Section 5.

4.5 Correlation between Sourcing Modes

As in Tomiura (2007), we look at the correlation between different international sourcing modes in Table 5. The upper panel (A) of this table shows the correlations at firm level. The inter-industry correlation between outsourcing and intra-firm imports is negative although insignificant. We find substituability between mixed strategies and outsourcing and also between mixed strategies and intra-firm imports.

	Intra-firm Trade	Outsourcing	Mixed Strategies	
	Panel	(A): At the firm	m level	
Intra-firm trade	1			
Outsourcing	-0.15	1		
Mixed transaction	-0.23*	-0.93***	1	
	Panel (B)	: At the transa	ction level	
Intra-firm trade	1			
Outsourcing	-0.84***	1		
Mixed transaction	-0.03	-0.52***	1	

Table 5 Correlation between sourcing modes (p-values into parentheses)

The correlation is for the share in the number of transaction across 239 industries ***, ** ,*, significant at 1%, 5% and 10% level of significance, respectively.

Panel (B) shows stronger inter-industry correlation at product level. The correlation between intra-firm imports and outsourcing is negative and significant at 1% level of significance. The correlation between outsourcing and mixed strategies is also significant and negative at 1% level of significance. The inter-industry correlations suggest a substitutability between the different modes of sourcing at product level. This finding is consistent with our theoretical model.

Overall, these cross-industry correlations suggest that the level of the firm might not be best suited for our empirical analysis since a firm may import (different) goods using different sourcing modes from different countries.

5 Results

5.1 Baseline Specification

Table 6 presents the estimates of the marginal effects of the regressions at transaction level. We evaluate the marginal effect at the sample means, which

measures the effect for a firm with characteristics equal to the sample averages. In the fourth specification (S1-S4), we estimate the model as close as possible to the theoretical framework.¹⁹ Note that all variables have been centered around their respective mean and that all specifications include a full set of French sector and product specific effects.

The first specification is an estimation of Equation 7. Except for the contracting environment whose coefficient has an unexpected sign, the marginal effects are in line with the theoretical predictions of the model and are significant at the 1% significance level.

Turning to the impact of the interaction terms on other covariate' effects, they do not qualitatively influence the results of the baseline regression. We use the Ai and Norton (2003) methodology to interpret the interaction effect whenever this effect varies widely across observations.

We show in specification (S1) that firms that have higher fixed costs choose to source their inputs through independent suppliers.²⁰ This is in line with our theoretical assumption. Consistent with this finding, we observe that firms that have a higher productivity level also have a higher probability of sourcing their input through independent suppliers. We find that a one percent increase in firm's TFP increases the probability of outsourcing by about a 0.028 percentage point, holding the other explanatory variables constant. These results are significant at a one percent level and robust across specifications.

Turning to the suppliers' input intensity, its marginal effect is positive and significant at 1%. Holding all other explanatory variables constant, we find that going from the lowest to the highest intensity in suppliers' input increases

¹⁹ Notice that the results remain robust to the exclusion of formal contractual relationships (alliances, franchising, joint ventures, licensing agreements). Only 1.5% of the transactions are formal outsourcing after the exclusion of mixed strategies. ²⁰ Notice that the results remain robust to the exclusion of the fixed costs and the primitive bargaining power variables.

	Label	(S1)	(S2)	(S3)	(S4)
Fixed Costs	f_i	0.034^{a}	0.034^{a}	0.034^{a}	0.034^{a}
		(0.002)	(0.002)	(0.002)	(0.002)
TFP	$ heta_i$	0.028^{a}	0.030^{a}	0.028^{a}	0.030^{a}
		(0.002)	(0.002)	(0.002)	(0.002)
Supplier's Input Intensity	z_i	0.307^{a}	0.242^{a}	0.306^{a}	0.242^{a}
		(0.024)	(0.025)	(0.024)	(0.025)
Contracting Environment	δ_l	-0.105^{a}	-0.103^{a}	-0.107^{a}	-0.105^{a}
		(0.014)	(0.014)	(0.014)	(0.014)
Primitive Bargaining Power	β_{o_i}	0.006^{a}	0.006^{a}	0.006^{a}	0.006^{a}
		(0.001)	(0.001)	(0.001)	(0.001)
Interaction term 1	$ heta_i imes z_i$		0.167^{a}		0.165^{a}
			(0.037)		(0.037)
Interaction term 2	$ heta_i imes \delta_l$			-0.034^{b}	-0.031^{b}
				(0.014)	(0.014)
French Sector Fixed Effects		Yes	Yes	Yes	Yes
Product Fixed Effects		Yes	Yes	Yes	Yes
Number of observation		62670	62670	62670	62670
Pseudo R^2		0.057	0.058	0.057	0.058
Log Likelihood		-33316	-33281	-33312	-33277

Table 6 Dependent variable: Y=1 for outsourcing (marginal effects presented.)

Robust standard error in brackets. a , b , c significantly different from 0 at 1%, 5% and 10% level, respectively.

the probability to outsource by 30.7 percentage points. This finding supports the first implication of the model. The probability that firms import their input from independent suppliers increases with the suppliers' input intensity of production.

In specification (S2), we introduce a multiplicative term between the suppliers' inputs intensity variable and the firm-level TFP variable in order to examine the second testable implication of the theoretical model. We find the mean interaction effect is significant and positive. This finding is robust across observations and confirms the theoretical prediction of the model. Greater intensity in suppliers' inputs increases the marginal effect of the TFP variable on the probability to source through an independent supplier. We find that a better contracting environment has a significant and negative impact on the probability of sourcing inputs through an independent supplier in specification (S1). Going from the lower to the upper bound of our "rule of law" variable decreases the probability to outsource by a 10.5 percentage point in the first specification. However, the theoretical model does not offer any predictions concerning the intercept-shift effect of the contracting environment variable on the sourcing mode. Looking at the mean interaction effect in specification (S3), we find that a higher productivity strengthens the negative marginal effect of the contracting environment on the probability of sourcing inputs through an independent supplier. Whereas the magnitude and the sign of the interaction effect are robust across observations, the significance level is not. The marginal effect of the interaction term is positive but insignificant for all observations that have larger predicted values than 0.8. These findings are not consistent with the third testable implication of the model but is in line with the prediction of a recent study by Antràs and Helpman (2008). According to the authors, the foreign supplier and the final good producer produce intermediate inputs that are partially contractible. They show that if a better contracting environment reduces the non-contractible share of the supplier's input then the incentives required by the final-good producer increases. Thus increasing the foreign contracting environment increases the likelihood of sourcing inputs through affiliates.²¹

Finally, we find that the primitive bargaining power measured by the logarithm of the size of the firm is also positive and significant at a 1% level in specification. Holding all explanatory variable constant, a one percent increase in the primitive bargaining power increases the probability of outsourcing by

²¹ Antràs and Helpman (2008) also show that a better contracting environment increases firms' entry. Considering our ranking of fixed costs, these new firms are less productive and import from affiliates.

about 0.006 percentage point.

5.2 Intermediate Inputs

We follow the methodology developed by Feenstra and Hanson (1996) to distinguish between final goods and imported intermediate inputs. For a sufficiently disaggregated level of sector classification, at HS3-level, we identify intermediate inputs as the products imported which are classified in an other sector than the sector in which the French firm reports its main activity. The results using imported intermediate inputs are reported in Table 7.

Table 7

Imported intermediate inputs' sample. Dependent variable: Y=1 for outsourcing (marginal effects presented.)

	Label	(I1)	(I2)	(I3)	(I4)
Fixed Costs	f_i	0.041^{a}	0.040^{a}	0.041^{a}	0.040^{a}
		(0.002)	(0.002)	(0.002)	(0.002)
TFP	$ heta_i$	0.027^{a}	0.030^{a}	0.027^{a}	0.030^{a}
		(0.002)	(0.002)	(0.002)	(0.002)
Supplier's Input Intensity	z_i	0.372^{a}	0.282^{a}	0.371^{a}	0.281^{a}
		(0.029)	(0.030)	(0.029)	(0.030)
Contracting Environment	δ_l	-0.074^{a}	-0.072^{a}	-0.076^{a}	-0.074^{a}
		(0.016)	(0.016)	(0.016)	(0.016)
Primitive Bargaining Power	β_{oi}	0.008^{a}	0.008^{a}	0.008^{a}	0.008^{a}
		(0.002)	(0.002)	(0.002)	(0.002)
Interaction term 1	$ heta_i imes z_i$		0.276^{a}		0.274^{a}
			(0.041)		(0.041)
Interaction term 2	$ heta_i imes \delta_l$			-0.024	-0.017
				(0.016)	(0.017)
Sector Fixed Effects		Yes	Yes	Yes	Yes
Product Fixed Effects		Yes	Yes	Yes	Yes
Number of observation		45829	45829	45829	45829
Pseudo R^2		0.064	0.067	0.064	0.067
Log Likelihood		-22731	-22670	-22729	-22670

Robust standard error into brackets. a , b , c significantly different from 0 at 1%, 5% and 10% level, respectively.

For both interaction terms, the sign of the marginal effects and the magni-

tude of the standard errors remain in line with the one found using the whole sample. Using the intermediate inputs sample slightly decreases the contracting environment's effect while it increases significantly the impact of the first interaction term. Notice that the Log likelihood is smaller when using the imported intermediate inputs sample. This is an indication that the sample of imported intermediate inputs fits the model better.

We find the first interaction terms positive and significant mean effects on the probability of sourcing inputs through an independent supplier. We find also that these results are robust across observations. This tends to confirm that the intermediate input sample is better suited for our analysis of the sourcing mode.

Our findings suggest that the contracting environment and firm's productivity level have a negative - albeit insignificant - mean interaction effect of the on the sourcing decision.

5.3 Factor Intensities and Factor Endowments

In Table 8, we consider the role of factor endowment and firm-level factor intensity in explaining the organizational choice. We show the results of estimations from the intermediate inputs sample in specifications (R1) to (R3).²²

In specification (R1), we find the countrys skill endowment and the firmspecific skill intensity to be important determinants of the organizational choice. The likelihood of sourcing intermediate inputs through an independent supplier decreases with the skill intensity of the firm's production process. This finding confirms earlier results at aggregate level by Antràs (2003),

 $^{^{22}\,\}rm We$ find similar results using the whole sample. These results are presented in Table C.1 of Appendix C.

		Intermediate Inputs			
	Label	(R1)	(R2)	(R3)	
Fixed Costs	f_i	0.040^{a}	0.091^{a}	0.090^{a}	
		(0.002)	(0.004)	(0.004)	
TFP	$ heta_i$	0.028^{a}	0.046^{a}	0.049^{a}	
		(0.002)	(0.002)	(0.003)	
Supplier's Input Intensity	z_i	0.394^{a}	0.372^{a}	0.281^{a}	
		(0.031)	(0.030)	(0.031)	
Contracting Environment	δ_l	-0.015	-0.184^{a}	-0.082^{a}	
		(0.021)	(0.023)	(0.023)	
Primitive Bargaining Power	β_{oi}	0.012^{a}	0.008^{a}	0.011^{a}	
		(0.002)	(0.002)	(0.002)	
Skill Endowment (Country)	H_l/L_l	-0.057^{a}		-0.053^{a}	
		(0.005)		(0.005)	
Skill Intensity (Firm)	s_i	-0.006^{a}		-0.005^{a}	
		(0.001)		(0.001)	
Capital-Labor Ratio (Country)	K_l/L_l		0.033^{a}	0.031^{a}	
			(0.005)	(0.005)	
Capital-Labor Ratio (Firm)	$(k/l)_i$		-0.065^{a}	-0.065^{a}	
			(0.004)	(0.004)	
Interaction term 1	$ heta_i imes z_i$			0.321^{a}	
				(0.045)	
Interaction term 2	$ heta_i imes \delta_l$			0.038^{c}	
				(0.020)	
Interaction term 3	$s_i \times H_l/L_l$	-0.014^{a}		-0.013^{a}	
		(0.003)		(0.002)	
Interaction term 4	$(k/l)_i \times K_l/L_l$		0.005	0.006^{c}	
			(0.004)	(0.004)	
Sector Fixed Effects		Yes	Yes	Yes	
Product Fixed Effects		Yes	Yes	Yes	
Number of observation		42123	42123	42123	
Pseudo R^2		0.071	0.075	0.083	
Log Likelihood		-20715	-20637	-20459	

Table 8 Robustness check: Dependent variable: Y=1 for outsourcing (marginal effects presented)

Robust standard error into brackets. a , b , c significantly different from 0 at 1%, 5% and 10% level, respectively.

Yeaple (2006) and Nunn and Trefler (2008). Moreover, the probability of importing inputs through independent suppliers is lower in countries that are relatively skill-abundant. The mean interaction term between the skill intensity and the skill endowment variable is negative and significant. The probability of sourcing inputs through affiliates is higher for high-skilled intensive firms that import from high-skilled abundant countries. The inclusion of the skill intensity and endowment variables sharply reduces the significance level of the marginal effect of the contracting environment variable. The role of the contracting environment is no longer important once we control for a country's specific skill endowment.

In specification (R2), we substitute the skill intensity and the skill endowment variables by the firm's and country's capital-labor ratio. The firm's capital intensity variable has a negative impact on the decision to source input through independent suppliers. This result is in line with Antràs (2003). We find however that the probability of sourcing through affiliates is higher, the higher the capital-labor ratio of the exporting country. In our database, the bulk of outsourcing is placed among European countries, USA and Canada. Among these countries, the largest recipients of outsourcing, Germany, Italy and Austria, have low skill endowments but high capital-labor ratios. The mean interaction term between the firm-level capital intensity and the country-level capital-labor ratio variables is positive and insignificant. We do not find any relationship between the capital intensities and the capital endowment.

6 Conclusion

Recent theoretical evidences have stressed the importance of a firm's productivity for their mode of sourcing foreign intermediate inputs. In Antràs and Helpman (2004), the most productive firms can compensate the foreign affiliate set-up costs, and source their intermediate inputs internally in the headquarter-intensive sector. We propose a slight modification of the Antràs and Helpman (2004) model by considering a production function that is firmspecific. The model predicts that the likelihood to import from an independent supplier is increasing with (i) the production intensity in supplier's inputs, (ii) the final-good producer's productivity and its production intensity in supplier's input, and (iii) its productivity and the contracting environment of the supplier's country.

We make several improvements to the previous empirical literature. First, we use firm-level data that have detailed information on the mode of sourcing at product-level and allow us to distinguish between final goods and intermediate inputs. Second, the data have sufficient information to compute total factor productivity at firm-level using Olley and Pakes (1996) correction for endogeneity of input selection. Third, we create a measure of intensity of supplier inputs at firm-level, which is derived from the theory.

The data indicate that French firms that source their intermediate inputs through independent suppliers have a higher fixed cost than firms that import intermediate inputs from their affiliates. The results further show that greater intensity in suppliers' inputs increases the likelihood of outsourcing. This result is in line with the prediction of the model. We find moreover that highly productive firms that use suppliers' inputs intensively in their production process have a higher probability to outsource. However, the data indicate that firms tend to import their inputs from their affiliates located in countries that have a better contracting environment. This might be explained by the level of contractibility of intermediate inputs as emphasized by Antràs and Helpman (2008). A careful examination of the contractibility hypothesis goes beyond the scope of our present analysis. Further research is needed to determine the exact nature of the relationship between contractibility and the sourcing mode of inputs.

References

- AI, C. and NORTON, E. C. (2003), "Interaction Terms in Logit and Probit Models", *Economics Letters*, vol. 80: pp. 123–129.
- ANTRÀS, P. (2003), "Firms, Contracts, and Trade Structure", Quarterly Journal of Economics, vol. 118 nº 4: pp. 1375–1418.
- ANTRÀS, P. (2005), "Incomplete Contracts and the Product Cycle", American Economic Review, vol. 95 nº 4: pp. 1054–1073.
- ANTRÀS, P. and HELPMAN, E. (2004), "Global Sourcing", *Journal of Political Economy*, vol. 112 nº 3: pp. 552–580.
- ANTRÀS, P. and HELPMAN, E. (2008), "Contractual Friction and Global Sourcing", in HELPMAN, E., MARIN, D. and VERDIER, T. (editors), *The Organization of Firms in a Global Economy*, Harvard University Press.
- BERNARD, A. B., JENSEN, J. B., REDDING, S. J. and SCHOTT, P. K. (2007), "Intra-Firm Trade", *mimeo*.
- CABRAL, L. M. B. and MATA, J. (2003), "On the Evolution of the Firm Size Distribution: Facts and Theory", American Economic Review, vol. 93 nº 4: pp. 1075–1090.
- FEENSTRA, R. C. and HANSON, G. H. (1996), "Globalization, Outsourcing, and Wage Inequality", American Economic Review, vol. 86 nº 2: pp. 240– 45.
- GROSSMAN, G. and HELPMAN, E. (2002), "Integration versus Outsourcing in Industry Equilibrium", *Quarterly Journal of Economics*, vol. 117: pp. 85–120.
- GROSSMAN, G. and HELPMAN, E. (2003), "Outsourcing versus FDI in Industry Equilibrium", Journal of the European Economic Association, vol. 1: pp. 317–327.
- GROSSMAN, G. and HELPMAN, E. (2005), "Outsourcing in the Global Econ-

omy", Review of Economic Studies, vol. 72 nº 1: pp. 135–159.

- GROSSMAN, G., HELPMAN, E. and SZEIDL, A. (2005), "Complementarities between Outsourcing and Foreign Sourcing", American Economic Review, Papers and Proceedings, vol. 95 nº 2: pp. 19–24.
- GROSSMAN, S. J. and HART, O. D. (1986), "The Costs and Benefits of Ownership: A Theory of Vertical and Lateral Integration", *Journal of Political Economy*, vol. 94 nº 4: pp. 691–719.
- HART, O. and MOORE, J. (1990), "Property Rights and the Nature of the Firm", *Journal of Political Economy*, vol. 98 nº 6: pp. 1119–1158.
- HELPMAN, E. (2006), "Trade, FDI, and the Organization of Firms", *Journal* of *Economic Literature*, vol. 44 n° 3: pp. 589–630.
- HELPMAN, E., MELITZ, M. and YEAPLE, S. (2004), "Export versus FDI with Heterogeneous Firms", American Economic Review, vol. 94: pp. 300– 316.
- KAUFMANN, D., KRAAY, A. and MASTRUZZI, M. (2006), "Governance Matters VI: Governance Indicators for 1996-2006", World Bank Working Paper, n^o 4280.
- MCLAREN, J. (2000), "Globalization and Vertical Structure", American Economic Review, vol. 90: pp. 1239–1254.
- MELITZ, M. (2003), "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity", *Econometrica*, vol. 71: pp. 1695– 1725.
- NORTON, E. C., WANG, H. and AI, C. (2004), "Computing Interaction Effects and Standard Errors in Logit and Probit Models", *Stata Journal*, vol. 4,2: pp. 154–167.
- NUNN, N. (2007), "Relationship-Specificity, Incomplete Contract, and The Pattern Of Trade", Quarterly Journal of Economics, vol. 122:2: pp. 569– 600.

- NUNN, N. and TREFLER, D. (2008), "The Boundaries of the Multinational Firm: An Empirical Analysis", in HELPMAN, E., MARIN, D. and VERDIER, T. (editors), *The Organization of Firms in a Global Economy*, Harvard University Press.
- OLLEY, S. and PAKES, A. (1996), "The Dynamics Of Productivity In The Telecommunications Equipment Industry", *Econometrica*, vol. 64 nº 6: pp. 1263–1298.
- PORTER, M. E. (1974), "Consumer Behavior, Retailer Power and Market Performance in Consumer Goods Industries", *The Review of Economics* and Statistics, vol. 56 nº 4: pp. 419–36.
- RASPILLER, S. and SILLARD, P. (2004), "Affiliating Versus Subcontracting: The Case of Multinationals", *INSEE*, Working Paper, vol. G-2004/9.
- SCHOTT, P. K. (2004), "Across-product Versus Within-product Specialization in International Trade", *The Quarterly Journal of Economics*, vol. 119 n^o 2: pp. 646–677.
- SPENCER, B. (2005), "International Outsourcing and Incomplete Contracts", Canadian Journal of Economics, vol. 38 nº 4: pp. 1107–1135.
- SUTTON, J. (1997), "Gibrat's Legacy", Journal of Economic Literature, vol. 35 nº 1: pp. 40–59.
- TOMIURA, E. (2005), "Foreign Outsourcing and Firm-level Characteristics: Evidence from Japanese Manufacturers", Journal of the Japanese and International Economies, vol. 19: pp. 255–271.
- TOMIURA, E. (2007), "Foreign Outsourcing, Exporting, and FDI: a Productivity Comparaison at the Firm Level", *Journal of International Economics*, vol. 72: pp. 113–127.
- TREFLER, D. (2002), "Increasing Returns to Scale and All That: A View From Trade", American Economic Review, vol. 92: pp. 93–119.

WILLIAMSON, O. E. (1985), The Economic Institutions of Capitalism, Free

Press.

YEAPLE, S. (2006), "Offshoring, Foreign Direct Investment, and the Structure of U.S. Trade", Journal of the European Economic Association, vol. 4 nº 2-3: pp. 602–611.

Appendices

A TFP Measurement

We use the Olley and Pakes (1996) (OP) semiparametric method to estimate firm-level TFP. This method allows robust estimation of the production function. It takes into account the endogeneity of some inputs, the exit of firms as well as the unobserved permanent differences among firms. The main assumption that the OP technique relies on, is the existence of a monotonic relationship between investment and firm-level unobserved heterogeneity. Estimation has been realized for each one of the 52 sectors (3 digit).

We consider the following Cobb-Douglas production function

$$Q_{it} = \lambda_0 + \lambda_K K_{it} + \lambda_L L_{it} + \lambda_M M_{it} + \theta_{it} + \epsilon_{it}$$

and denote the logarithm of output, capital, labor and intermediate inputs with Q_{it} , K_{it} , L_{it} M_{it} , respectively. Subscripts *i* and *t* stand for firm and time, θ_{it} denotes productivity, and ϵ_{it} stands for measurement error in output. It is assumed that θ_{it} follows an exogenous first order Markov process:

$$\theta_{it+1} = E[\theta_{it+1}|\theta_t] + \upsilon_{it+1}$$

where v_{it} is uncorrelated with the productivity shock. The endogeneity problem stems from the fact that K_{it} and L_{it} are correlated with the θ_{it} . This makes λ_{OLS} to be biased and inconsistent. Given that investment is strictly monotonic, it can be inverted as:

$$\theta_{it} = h(I_{it}, K_{it})$$

and substituting this function in the production function leads to

$$Q_{it} = \lambda_L L_{it} + \lambda_M M_{it} + \Phi(I_{it}, K_{it}) + \epsilon_{it}$$

where $\Phi(I_{it}, K_{it}) = \lambda_0 + \lambda_K K_{it} + h(I_{it}, K_{it})$. Since the functional form of $\Phi(\cdot)$ is not known, we cannot estimate the coefficients of the capital and labor variable directly. Instead, we use a linear model that includes a series estimator using a full interaction term polynomial in capital and investment to approximate $\Phi(\cdot)$. From this first stage, the consistent estimates of the coefficients on labor and material inputs as well as the estimate of the polynomial in I_{it} and K_{it} are obtained.

The second stage takes into account the survival of firms. These probabilities are given by:

$$Pr\{\chi_{t+1} = 1 | \underline{\theta}_{t+1}(K_{t+1}), Jt\} = Pr\{\theta_{t+1} \ge \underline{\theta}_{t+1}(K_{t+1}) | \underline{\theta}_{t+1}(K_{t+1}), \theta_t\}$$
$$= \varphi\{\underline{\theta}_{t+1}(K_{t+1}), \theta_t\}$$
$$= \varphi(i_t, K_t)$$
$$= P_t$$

Where the probability that a firm survives at time t + 1 is conditional on its information set at time t, J_t and θ_{t+1} . This is equal to the probability that the firm's productivity is greater than a threshold, $\underline{\theta}_{t+1}$, which in turn depends on the capital stock. The survival probability can be written as a function of investment and capital stock at time t. Thus, we estimate a probit regression on a polynomial in investment and capital controlling for year specific effects. Now, consider the expectation $Q_{t+1} - \lambda_L L_{t+1}$ conditional on the information at time t and survival at t + 1.

$$E[Q_{t+1} - \lambda_L L_{t+1} | K_{t+1}, \chi_{t+1} = 1] = \lambda_0 + \lambda_K K_{t+1} + E[\theta_{t+1} | \theta_t, \chi_{t+1} = 1]$$

= $\lambda_K K_{t+1} + g(\underline{\theta}_{t+1}, \theta_t)$

 θ_{it} follows an exogenous first-order Markov process. We substitute the productivity shock in the above equation using the result from the first stage.

$$Q_{t+1} - \lambda_L L_{t+1} = \lambda_K K_{t+1} + g(P_t, \Phi_t - \lambda_K K_t) + v_{t+1} + \epsilon_{it}$$

The third step takes the estimates from λ_L , Φ_t , and P_t and substitutes them for the true values. The series estimator is obtained by running a non-linear least squares on the equation

$$Q_{t+1} - \lambda_L L_{t+1} - \lambda_M M_{t+1} = c + \lambda_K K_{t+1} + \sum_{j=0}^{s-M} \sum_{M=0}^{s} \lambda_{Mj} (\hat{\phi}_t - \lambda_K K_t)^M \hat{P}_t^j + e_t$$

where s is the order of the polynomial used to estimate the coefficient on capital.

B Descriptive Statistics and Correlation Table

	Label	Mean	Std. Dev.	Obs.
Fixed Costs (Log)	f_i	-1.116	0.917	62670
TFP (Log)	$ heta_i$	0.003	1.066	62721
Supplier's Input Intensity	z_i	0.001	0.097	62721
Firm Size (Log)	β_{oi}	-0.002	0.132	62721
Contracting Environment	δ_l	-0.085	1.414	62721
Country Skill Endowment (Log)	H_l/L_l	0.001	2.294	56912
Firm Skill Intensity (Log)	s_i	0.001	0.432	56912
Country Capital-Labor Ratio (Log)	K_l/L_l	0.001	0.966	56912
Firm Capital-Labor Ratio (Log)	$(k/l)_i$	0.001	0.524	56912
Interaction term 1	$z_i imes heta_i$	0.026	0.115	62721
Interaction term 2	$ heta_i imes \delta_l$	-0.005	0.140	62721
Interaction term 3	$s_i \times H_l/L_l$	0.043	0.942	56912
Interaction term 4	$(k/l)_i \times K_l/L_l$	0.053	0.592	56912

Table B.1 Summary statistics of variables

Non-interacted variables have been centered around their mean.

Table B.2 Correlation matrix

	f_i	$ heta_i$	z_i	β_0	δ_l	s_i	H_l/L_l	$(k/l)_i$	K_l/L_l
f_i	1.000								
$ heta_i$	-0.164^{a}	1.000							
z_i	-0.205^{a}	0.253^{a}	1.000						
β_{0i}	0.126^{a}	0.198^{a}	-0.005	1.000					
δ_l	0.100^{a}	-0.039^{a}	-0.036^{a}	-0.016^{a}	1.000				
s_i	0.006^{a}	0.059^{a}	0.091^{a}	0.236^{a}	0.005^{a}	1.000			
H_l/L_l	0.016^{a}	0.013^{a}	0.011^{a}	0.059^{a}	0.454^{a}	0.044^{a}	1.000		
$(k/l)_i$	0.765^{a}	0.071^{a}	-0.152^{a}	0.150^{a}	0.104^{a}	0.076^{a}	0.050^{a}	1.000	
K_l/L_l	0.108^{a}	-0.067^{a}	-0.044^{a}	-0.021^{a}	0.584^{a}	-0.014^{a}	0.319^{a}	0.104^{a}	1.000
$ \begin{array}{c} \Pi_l/L_l\\(k/l)_i\\K_l/L_l \end{array} $	0.010^{a} 0.765^{a} 0.108^{a}	0.071^{a} - 0.067^{a}	-0.152^{a} -0.044^{a}	0.059^{a} 0.150^{a} -0.021^{a}	0.434^{a} 0.104^{a} 0.584^{a}	0.076^{a} - 0.014^{a}	0.050^{a} 0.319^{a}	$1.000 \\ 0.104^{a}$	1.000

 a, significantly different from 0 at 1% level.

Robustness Checks: Factor Intensities and Factor Endowments using the Whole Sample \mathbf{C}

Table C.1 Robustness check: Dependent variable: Y=1 for outsourcing (marginal effects presented)

		V	Whole Sample				
	Label	(R4)	(R5)	(R6)			
Fixed Costs	f_i	0.034^{a}	0.095^{a}	0.096^{a}			
		(0.002)	(0.003)	(0.004)			
TFP	$ heta_i$	0.028^{a}	0.049^{a}	0.051^{a}			
		(0.002)	(0.002)	(0.002)			
Supplier's Input Intensity	z_i	0.326^{a}	0.318^{a}	0.241^{a}			
		(0.026)	(0.025)	(0.026)			
Contracting Environment	δ_l	-0.035^{c}	-0.195^{a}	-0.106^{a}			
		(0.018)	(0.020)	(0.021)			
Primitive Bargaining Power	β_{oi}	0.010^{a}	0.006^{a}	0.008^{a}			
		(0.002)	(0.001)	(0.001)			
Skill Endowment (Country)	H_l/L_l	-0.056^{a}		-0.052^{a}			
		(0.005)		(0.005)			
Skill Intensity (Firm)	s_i	-0.006^{a}		-0.004^{a}			
		(0.001)		(0.001)			
Capital-Labor Ratio (Country)	K_l/L_l		0.029^{a}	0.032^{a}			
			(0.004)	(0.004)			
Capital-Labor Ratio (Firm)	$(k/l)_i$		-0.078^{a}	-0.078^{a}			
			(0.003)	(0.003)			
Interaction term 1	$z_i imes heta_i$			0.221^{a}			
				(0.040)			
Interaction term 2	$ heta_i imes \delta_l$			0.017			
				(0.017)			
Interaction term 3	$s_i \times H_l/L_l$	-0.010^{a}		-0.010^{a}			
		(0.002)		(0.002)			
Interaction term 4	$(k/l)_i \times K_l/L_l$		0.020^{a}	0.020^{a}			
			(0.004)	(0.004)			
Sector Fixed Effects		Yes	Yes	Yes			
Products Fixed Effects		Yes	Yes	Yes			
Number of observation		56912	56912	56912			
Pseudo R^2		0.061	0.068	0.072			
Log Likelihood		-30161	-29967	-29809			

Robust standard error into brackets. a, b, c significantly different from 0 at 1%, 5% and 10% level, respectively.

D Results from aggregate level data

We follow Yeaple (2006) and Bernard *et al.* (2007) and use aggregate data on intra-firm trade between French parent firms and their foreign affiliate to measure FDI by French multinational firms. The data comes from the 1999 SESSI survey. For each French affiliate of a French multinational firm, we observe (1) its country of residence, (2) the value of its exports to its parent firm (3) the type of products it exports (at 3-digit level) and (4) the manufacturing sector of its parent firm. Since our primary focus is on industry variation, our measure of intra-firm trade is the share of intra-firm imports in total imports by French manufacturing sectors, products and origin countries. Total imports has been computed as the sum of imports from affiliates plus imports from independent suppliers from the SESSI survey.²³ The dataset contains 52 manufacturing sectors, 67 imported products and 135 countries.

We measure the extent of dispersion across firms within an industry using two measures of productivity dispersion. We calculate first, firm's total factor productivity (TFP) using the Olley and Pakes (1996) methodology and compute for each French manufacturing sector the standard deviation of the logarithm of firm TFP across firms within an industry. Second, we follow Yeaple (2006) and compute the standard deviation of the logarithm of firm sale across firms within an industry. The differences in firm size as measured by final good sales reflect differences in productivity because more productive firms should sell more. We estimate the model using an OLS and a Poisson regressions. we control for product and country specific heterogeneity by using a full set of products and country fixed effects. The Poisson method of estimation has the advantage of avoiding to take the logarithm of zero values (? and ?).

 $^{^{23}}$ For each combination of manufacturing sector and country, we aggregate imports at the 3-digit product level.

Dispersion Measure	OLS	Poisson	OLS	Poisson
Productivity	-9.493^{a}	-0.341^{a}		
	(3.198)	(0.113)		
Sales			-2.463^{c}	-0.085^{c}
			(1.314)	(0.045)
Country fixed effects	Yes	Yes	Yes	Yes
Product fixed effects	Yes	Yes	Yes	Yes
Observations	13593	13593	13593	13593
Positive observations	7115	7115	7115	7115
R-squared	0.07	0.07	0.07	0.07
Log likelihood		-362598.26		-362733.29

Table D.1 Dependent variable: share of intra-firm trade

Robust standard error into brackets. a , b , c significantly different from 0 at 1%, 5% and 10% level, respectively.

Our results suggests that intra-firm import is lower in industry with higher dispersion in productivity. This results is robust to the measure of dispersion and to the econometric methodology. The results is consistent with the prediction of Antràs and Helpman (2004), but for a different ranking of fixed costs.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

841	Barbara Petrongolo	What are the Long-Term Effects of UI? Evidence from the UK JSA Reform
840	Kevin D. Sheedy	Robustly Optimal Monetary Policy
839	Christopher A. Pissarides	The Unemployment Volatility Puzzle: Is Wage Stickiness the Answer?
838	Kevin D. Sheedy	Inflation Persistence when Price Stickiness Differs between Industries
837	Kevin D. Sheedy	Intrinsic Inflation Persistence
836	Andrew E. Clark Ed Diener Yannis Georgellis Richard E. Lucas	Lags and Leads in Life Satisfaction: A Test of the Baseline Hypothesis
835	Jean-Baptiste Michau	Creative Destruction with On-the-Job Search
834	Nikolaus Wolf	Scylla and Charybodis: the European Economy and Poland's Adherence to Gold, 1928-1936
833	Sharon Belenzon Tomer Berkovitz	Innovation in Business Groups
832	Espen R. Moen Åsa Rosén	Incentives in Competitive Search Equilibrium
831	Eran Yashiv	U.S. Labor Market Dynamics Revisited
830	Laura Alfaro Andrew Charlton	Growth and the Quality of Foreign Direct Investment: Is All FDI Equal?
829	Richard Layard David Clark Martin Knapp Guy Mayraz	Cost-Benefit Analysis of Psychological Therapy
828	Emma Tominey	Maternal Smoking During Pregnancy and Early Child Outcomes
827	Christos Genakos Tommaso Valletti	Testing the "Waterbed" Effect in Mobile Telephony

826	Luis Garicano Paul Heaton	Information Technology, Organization, and Productivity in the Public Sector: Evidence from Police Departments
825	Laura Alfaro Andrew Charlton	Intra-Industry Foreign Direct Investment
824	Miklós Koren Silvana Tenreyro	Technological Diversification
823	Régis Barnichon	The Shimer Puzzle and the Correct Identification of Productivity Shocks
822	Carlos Thomas	Search Frictions, Real Rigidities and Inflation Dynamics
821	Giulia Faggio Kjell G. Salvanes John Van Reenen	The Evolution of Inequality in Productivity and Wages: Panel Data Evidence
820	Luis Araujo Emanuel Ornelas	Trust-Based Trade
819	Régis Barnichon	Productivity, Aggregate Demand and Unemployment Fluctuations
818	Rachel Griffith Sokbae Lee John Van Reenen	Is Distance Dying at Last? Falling Home Bias in Fixed Effects Models of Patent Citations
817	Tsutomu Miyagawa YoungGak Kim	Measuring Organization Capital in Japan: An Empirical Assessment Using Firm-Level Data
816	Andreas Georgiadis Alan Manning	Spend It Like Beckham? Inequality and Redistribution in the UK, 1983-2004
815	Guy Michaels Xiaojia Zhi	Freedom Fries
814	Ralph Ossa	Trade Liberalization, Outsourcing, and Firm Productivity
813	Karsten Kohn Alexander C. Lembcke	Wage Distributions by Bargaining Regime: Linked Employer-Employee Data Evidence from Germany
812	Yannis M. Ioannides Henry G. Overman Esteban Rossi-Hansberg Kurt Schmidheiny	The Effect of Information and Communication Technologies on Urban Structure
811	Guy Michaels	The Division of Labor, Coordination, and the Demand for Information Processing

The Centre for Economic Performance Publications Unit Tel 020 7955 7673 Fax 020 7955 7595 Email <u>info@cep.lse.ac.uk</u> Web site http://cep.lse.ac.uk